Author’s response to reviews

Title: Comparison of M-Mode Echocardiographic Left Ventricular Mass Measured Using Digital and Strip Chart Readings: The Atherosclerosis Risk in Communities (ARIC) Study

Authors:

Dr Donna K Arnett (arnett@epi.umn.edu)
Thomas N Skelton (tskelton@medicine.umsmed.edu)
Philip R Liebson (pliebson@rpslmc.edu)
Emelia Benjamin (emelia@bu.edu)
Richard G Hutchinson (rhutchinson@medicine.umsmed.edu)

Version: 2 Date: 22 May 2003

April 28, 2003

Cardiovascular Ultrasound
Online Resubmission

RE: Resubmission of "Comparison of m-mode echocardiographic left ventricular mass measured using digital and strip chart readings: The Atherosclerosis Risk in Communities (ARIC) Study.

Please accept our online resubmission of the above referenced manuscript. We have addressed the reviewer's concerns and have 1) deleted figure 2; 2) fixed reference #12 as suggested; 3) added an introduction; and 4) deleted references from the abstract.

If you have any questions or concerns, please contact me by phone 612/626-8863, fax 612/624-0315, or email at arnett@epi.umn.edu.

Sincerely,

Donna K. Arnett, PhD
Associate Professor